Author’s response to reviews

Title: Reviewer agreement trends from four years of electronic submissions of conference abstracts.

Authors:

Brian H Rowe (brian.rowe@ualberta.ca)
Carol Spooner (cspooner@ualberta.ca)
Sandra Blitz (SandraBlitz@cha.ab.ca)
Trevor L Strome (tstrom@wrha.mb.ca)
Eric Grafstein (egrafstein@providencehealth.bc.ca)
Andrew Worster (aworster@rogers.com)

Version: 4 Date: 10 February 2006

Author’s response to reviews: see over
February 9, 2006

The BioMed Central Editorial Team

*BioMed Central*
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: editorial@biomedcentral.com
Web: [http://www.biomedcentral.com/](http://www.biomedcentral.com/)

To whom it may concern,

RE: “Reviewer agreement trends from four years of electronic submissions of conference abstracts.” for review by *BioMed Central*

Thank you for your letter dated February 7, 2006 regarding the decision by the *BioMed Central* to accept the above titled manuscript. We have reviewed the minor comments provided by two reviewers and are submitting the following final revision of the manuscript to the Editors. The reviewer comments were specifically addressed in the attached COMMENTS LETTER and changes made in the corresponding sections of the manuscript. All the changes in the manuscript are highlighted in **bold** print.

We would like to thank *BioMed Central* and the reviewers for their continued feedback on this manuscript. We hope that the changes we have made to the manuscript and the response to the reviewer’s questions complete this review process.

Respectfully submitted,

Brian H. Rowe, MD, MSc, CCFP(EM), FCCP
Canada Research Chair in Emergency Airway Diseases
Professor and Research Director, Department of Emergency Medicine
University of Alberta